Author’s response to reviews

Title: Unusual presentation of a giant thoracic schwannoma as an acute abdomen

Authors:

Isaac Yang MD (yangi@neurosurg.ucsf.edu)
Elena Paik MD (paike@uci.edu)
Nancy G Huh MD (nanhuh1@yahoo.com)
Andrew T Parsa MD (parsaA@neurosurg.ucsf.edu)
Christopher P Ames MD (AmesC@neurosurg.ucsf.edu)

Version: 2 Date: 12 August 2008

Author’s response to reviews:

We would like to submit a Case Report on a very unusual presentation of a large Giant thoracic schwannoma that presented as an acute abdomen. The patient underwent an emergent laparotomy, appendectomy, and several Endoscopic GI studies before isolating this giant schwannoma. A brief, but complete literature review is also included.

We have made the appropriate format as needed by the journal. We have cited the ethnicity, the consent, limited the number of figures and references.

Hopefully this manuscript would make a good publication for the Journal of Medical Case Reports. We appreciate your time and consideration.